

GOPEN ACCESS

Citation: Galileya LT, Wasmann RE, Chabala C, Rabie H, Lee J, Njahira Mukui I, et al. (2023) Evaluating pediatric tuberculosis dosing guidelines: A model-based individual data pooled analysis. PLoS Med 20(11): e1004303. https://doi.org/ 10.1371/journal.pmed.1004303

Academic Editor: Marie-Louise Newell, University of Southampton, UNITED KINGDOM

Received: November 17, 2022

Accepted: October 2, 2023

Published: November 21, 2023

Copyright: © 2023 Galileya et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: The data underlying the results of this study are available upon request because they contain potentially sensitive information. Interested researchers may contact the Drugs for Neglected Diseases initiative (DNDi), commissioner of part of the data of this study, for data access requests via email at CTdata@dndi.org. Researchers may also request data by completing the form available at https://www.dndi.org/ category/clinical-trials/. In this, they confirm that they will share data and results with DNDi and will publish any results open access. The SHINE data RESEARCH ARTICLE

Evaluating pediatric tuberculosis dosing guidelines: A model-based individual data pooled analysis

Lufina Tsirizani Galileya^{1,2}, Roeland E. Wasmann¹, Chishala Chabala^{1,3,4}, Helena Rabie⁵, Janice Lee⁶, Irene Njahira Mukui⁶, Anneke Hesseling⁷, Heather Zar⁸, Rob Aarnoutse⁹, Anna Turkova¹⁰, Diana Gibb¹⁰, Mark F. Cotton⁵, Helen McIlleron^{1,11}, Paolo Denti^{1*}

 Division of Clinical Pharmacology, Department of Medicine, University of Cape Town, Cape Town, South Africa, 2 Training and Research Unit of Excellence, Kamuzu University of Health Sciences, Blantyre, Malawi,
Department of Pediatrics, University of Zambia, School of Medicine, Lusaka, Zambia, 4 University Teaching Hospitals-Children's Hospital, Lusaka, Zambia, 5 Department of Pediatrics and Child Health and Family Center for Research with Ubuntu, Stellenbosch University, Cape Town, South Africa, 6 Drugs for Neglected Diseases initiative, Geneva, Switzerland, 7 Desmond Tutu TB Centre, Department of Pediatrics and Child Health, Faculty of Medicine and Health Sciences, Stellenbosch University, Cape Town, South Africa, 8 Department of Pediatrics and Child Health, Red Cross War Memorial Children's Hospital, and SA-MRC Unit on Child & Adolescent Health, University of Cape Town, Cape Town, South Africa, 9 Radboud University Medical Center, Nijmegen, the Netherlands, 10 Medical Research Council Clinical Trials Unit at University College London, Institute of Clinical Trials and Methodology, London, United Kingdom,
Wellcome Centre for Infectious Diseases Research in Africa (CIDRI-Africa), Institute of Infectious Disease and Molecular Medicine, University of Cape Town, South Africa

* paolo.denti@uct.ac.za

Abstract

Background

The current World Health Organization (WHO) pediatric tuberculosis dosing guidelines lead to suboptimal drug exposures. Identifying factors altering the exposure of these drugs in children is essential for dose optimization. Pediatric pharmacokinetic studies are usually small, leading to high variability and uncertainty in pharmacokinetic results between studies. We pooled data from large pharmacokinetic studies to identify key covariates influencing drug exposure to optimize tuberculosis dosing in children.

Methods and findings

We used nonlinear mixed-effects modeling to characterize the pharmacokinetics of rifampicin, isoniazid, and pyrazinamide, and investigated the association of human immunodeficiency virus (HIV), antiretroviral therapy (ART), drug formulation, age, and body size with their pharmacokinetics.

Data from 387 children from South Africa, Zambia, Malawi, and India were available for analysis; 47% were female and 39% living with HIV (95% on ART). Median (range) age was 2.2 (0.2 to 15.0) years and weight 10.9 (3.2 to 59.3) kg. Body size (allometry) was used to scale clearance and volume of distribution of all 3 drugs. Age affected the bioavailability of rifampicin and isoniazid; at birth, children had 48.9% (95% confidence interval (CI) [36.0%,

are held at Medical Research Council Clinical Trials Unit at University College London, which encourages optimal use of data by employing a controlled access approach to data sharing, incorporating a transparent and robust system to review requests and provide secure data access consistent with the relevant ethics committee approvals. We will consider all requests for data sharing, which can be initiated by contacting mrcctu.ctuenquiries@ucl.ac.uk. The DATiC data are stored on University of Cape Town's institutional data repository (ZivaHub) under the following doi:10.25375/uct.22930607. We confirm that other researchers who meet the criteria for access to confidential data will be able to access these data in the same manner as the authors.

Funding: The author(s) received no specific funding for this work.

Competing interests: The authors have declared that no competing interests exist.

Abbreviations: ART, antiretroviral therapy; BOV, between-occasion variability; BSV, between-subject variability; CI, confidence interval; DNDi, Drugs for Neglected Diseases initiative; FDC, fixed-dose combination; FFM, fat-free mass; FOCE-I, firstorder conditional estimation with interaction; HIV, human immunodeficiency virus; LLOQ, lower limit of quantification; LOD, limit of detection; NAT2, Nacetyltransferase 2; SAEM, stochastic approximation expectation-maximization; SIR, sampling importance resampling; UCT, University of Cape Town; VPC, visual predictive check; WHO, World Health Organization. 61.8%]; p < 0.001) and 64.5% (95% CI [52.1%, 78.9%]; p < 0.001) of adult rifampicin and isoniazid bioavailability, respectively, and reached full adult bioavailability after 2 years of age for both drugs. Age also affected the clearance of all drugs (maturation), children reached 50% adult drug clearing capacity at around 3 months after birth and neared full maturation around 3 years of age. While HIV per se did not affect the pharmacokinetics of firstline tuberculosis drugs, rifampicin clearance was 22% lower (95% CI [13%, 28%]; p < 0.001) and pyrazinamide clearance was 49% higher (95% CI [39%, 57%]; p < 0.001) in children on lopinavir/ritonavir; isoniazid bioavailability was reduced by 39% (95% CI [32%, 45%]; p < 0.001) when simultaneously coadministered with lopinavir/ritonavir and was 37% lower (95% CI [22%, 52%]; p < 0.001) in children on efavirenz. Simulations of 2010 WHO-recommended pediatric tuberculosis doses revealed that, compared to adult values, rifampicin exposures are lower in most children, except those younger than 3 months, who experience relatively higher exposure for all drugs, due to immature clearance. Increasing the rifampicin doses in children older than 3 months by 75 mg for children weighing <25 kg and 150 mg for children weighing >25 kg could improve rifampicin exposures. Our analysis was limited by the differences in availability of covariates among the pooled studies.

Conclusions

Children older than 3 months have lower rifampicin exposures than adults and increasing their dose by 75 or 150 mg could improve therapy. Altered exposures in children with HIV is most likely caused by concomitant ART and not HIV per se. The importance of the drug–drug interactions with lopinavir/ritonavir and efavirenz should be evaluated further and considered in future dosing guidance.

Trial registration

ClinicalTrials.gov registration numbers; NCT02348177, NCT01637558, ISRCTN63579542

Author summary

Why was this study done?

- Recent pharmacokinetic studies in children on 2010 World Health Organization (WHO)-recommended first-line tuberculosis treatment found variable exposures of rifampicin, isoniazid, and pyrazinamide, with rifampicin being the drug achieving the lowest exposures.
- Findings of the association of human immunodeficiency virus (HIV) with the exposures of first-line tuberculosis drugs are inconsistent, and the relationships of HIV, antiretro-virals, formulation, demographics, and different sample processing and analysis methods are unclear.
- To rapidly mitigate the low rifampicin exposures, policymakers have been looking to investigate if taking the existing fixed dose combination tablet along with an additional 75 mg or 150 mg tablet of rifampicin alone (top-up) would be a good way to optimize rifampicin doses in children.

What did the researchers do and find?

- We pooled individual pharmacokinetic data from 3 large studies, which gave us a total of 387 children through whom we investigated the association of HIV, antiretrovirals, and formulation, and investigated the exposures achieved after increasing rifampicin doses by 75 mg or 150 mg.
- Age affected drug bioavailability and clearance and there was an association between body size and both the volume of distribution and drug clearance of first-line tuberculosis drugs.
- HIV had no association with the exposures of first-line tuberculosis drugs; rather, the antiretroviral lopinavir/ritonavir increased rifampicin exposure and reduced isoniazid and pyrazinamide exposure.
- Increasing rifampicin doses can lead to rifampicin exposures comparable to those observed in adults, and we report large differences between rifampicin and isoniazid formulations with reduced exposures in some formulations.

What do these findings mean?

- There are opportunities to rapidly improve first-line pediatric tuberculosis treatment through increasing rifampicin doses, which can be done through a rifampicin tablet top-up to the current fixed dose combination tablet as part of the WHO guidelines.
- Future studies investigating the association of HIV with tuberculosis drug exposures should explore the relationship of different antiretroviral therapies with tuberculosisdrug exposures, including drug-drug interactions due to administration of the drugs at the same time.
- Where possible, children on HIV and tuberculosis co-treatment should be closely monitored for possible toxicities or reduced efficacy arising from drug-drug interactions.
- The main limitation of our study is the small study design differences among the pooled studies, which led to some unattributable associations that we classified as site or study associations.

Introduction

Children account for 11% of global tuberculosis cases [1]. Approximately 1.1 million children develop tuberculosis disease and 210,000 die of tuberculosis complications each year [1,2]. Human immunodeficiency virus (HIV) increases the risk for active tuberculosis disease 20-fold. HIV and tuberculosis coinfection and co-medication are, therefore, common [3].

Diagnosing and treating tuberculosis in children is challenging because of the limited diagnostics and restricted treatment options in this population. The World Health Organization (WHO) recommends first-line tuberculosis treatment for children, comprising rifampicin 15 (10 to 20) mg/kg, isoniazid 10 (7 to 15) mg/kg, and pyrazinamide 35 (30 to 40) mg/kg with or without ethambutol 20 (15 to 25) mg/kg. This regimen is given to children with drug-susceptible tuberculosis for 2 months, followed by 4 months of rifampicin and isoniazid [4]. Because of potential drug-drug and drug-disease interactions, children with HIV and tuberculosis coinfection are at risk of suboptimal exposures of tuberculosis drugs [5–7].

Previous studies have reported contradictory findings on the association of HIV on the pharmacokinetics of tuberculosis drugs in both children and adults [5]. Some antiretroviral therapies (ART) have been reported to alter the exposure of tuberculosis drugs in adults. For example, efavirenz has been reported to increase isoniazid clearance by 24%, thereby reducing its exposure [6]. Ritonavir was reported to decrease rifampicin clearance by 46%, increasing its exposure [7]. There is a paucity of data to support dose adjustments to negate the associations of HIV and ART on the exposure of first-line tuberculosis drugs in children. The existing studies are small, probably due to the ethical regulations surrounding recruiting children in pharmacokinetic clinical trials and the high cost of conducting them [8].

Reduced antituberculosis drug exposures, which may result from, for example, faulty formulations, drug–drug interactions, or not accounting for age and body size, may reduce efficacy, leading to treatment failure or relapse, with an increased risk of drug resistance requiring longer, complex, and more expensive treatment [9]. On the other hand, overexposure increases the risk of toxicity [10]. It is therefore essential that childhood tuberculosis dosing is optimized for all children, including those coinfected with HIV.

Understanding and quantifying the associations of HIV, ART, drug formulation, age, and body size on the pharmacokinetics of tuberculosis drugs is essential to optimizing treatment in children with tuberculosis and HIV coinfection. However, conducting clinical trials to answer these questions is challenging and expensive [8]. Pooling datasets of existing pediatric clinical studies is therefore a good approach to answering these questions and can increase the power to find relationships that smaller studies cannot. We conducted a pooled pharmacokinetic analysis of individual pediatric patient data from 3 multisite studies conducted in South Africa, Malawi, India, and Zambia. We aimed to characterize the pharmacokinetics of first-line tuberculosis drugs, quantify the association of HIV, ART, age, body size, and drug formulation on the exposures of these drugs, and evaluate the current WHO-recommended pediatric tuberculosis dosing guidelines in children.

Methods

Study design and setting

The first study included the pooled analysis was SHINE (Clinical trial registration number ISRCTN63579542), a randomized controlled trial of therapy shortening for minimal tuberculosis with 2010 WHO-recommended doses using fixed-dose combination (FDC) drugs in African and Indian children with or without HIV. Children were randomized to either 4 or 6 months (2 months intensive phase in each case) first-line tuberculosis treatment [11].

The second study, conducted by Drugs for Neglected Diseases initiative (DNDi) (Clinical trial registration number NCT02348177), was a Phase I/II open-label, sequential non-randomized study comparing pharmacokinetics of lopinavir boosted with ritonavir (4:1 ratio) without rifampicin versus lopinavir super-boosted with ritonavir (4:4 ratio) in the presence of rifampicin in children [12].

DATiC (Clinical trial registration number NCT01637558) was a pharmacokinetic study investigating optimal dosing of first-line antituberculosis and ART drugs in children. On the day of pharmacokinetic sampling, single (not co-formulated) tuberculosis-drug formulations were used in doses according to WHO 2010 guidelines. Before and after pharmacokinetic

Study	Study period	Sites	Age*	Weight*	HIV status*	Health condition*	TB-drug formulations	Pharmacokinetic sampling day(s)	Pharmacokinetic sampling schedule
SHINE [11]	July 2016 to July 2018	i) Cape Town, South Africa ii) Lusaka, Zambia iii) Chennai & Pune in India	0–16 years	≥3 kg	HIV + and HIV-	Children with minimal TB**	Pediatric and adult FDC from Macleod's Pharmaceuticals Ltd	i) HIV-: 2–8 weeks after starting treatment ii) HIV+: during the last month of continuation phase of TB treatment	Pre-dose, 1, 2, 4, 6, 8, and 12 hours post-dose
DNDi [12]	January 2013 to November 2015	i) Johannesburg ii) Cape Town iii) Durban South Africa	>42 weeks gestational age	3–15 kg	HIV+	Children with TB	Pediatric FDC of rifampicin and isoniazid from Sandoz Pharmaceuticals Ltd Single pediatric pyrazinamide tablet from Sandoz Pharmaceuticals Ltd	During second month of co-treatment with rifampicin, during the last month of co-treatment and 4–6 weeks after rifampicin treatment	Pre-dose, 1, 2, 4, 6, and 10 hours post- dose
DATIC [13]	November 2012 to June 2017	i) Blantyre, Malawi ii) Cape Town, South Africa	<12 years	1.5–30 kg	HIV + and HIV-	Children with TB (those with acute severe illness excluded)	On day of pharmacokinetic sampling; Rifampicin: i) Eremfat, Riemser Arzneimittel, Germany ii) R-Cin Aspen Pharmacare, South Africa iii) Rimactazid, Novartis, India. Isoniazid: i) Riemser Arzneimittel, Germany ii) Rimactazid, Novartis, India. Pyrazinamide: i) Svizera Laboratories. India	At least 2 weeks after starting intensive TB treatment phase	Pre-dose, 1, 2, 4, 6, 8, and 10 hours post-dose

Table 1.	Characteristics	of studies	included	in the	pooled	analysis.
----------	-----------------	------------	----------	--------	--------	-----------

*Study inclusion criteria.

**Defined as respiratory tuberculosis confined to 1 lobe (opacification of <1 lobe) with no cavities, no signs of miliary tuberculosis, no complex pleural effusion, and no clinically significant airway obstruction or peripheral lymph-node tuberculosis.

DNDi, Drugs for Neglected Diseases initiative; FDC, fixed-dose combination; HIV, human immunodeficiency virus; TB, tuberculosis.

https://doi.org/10.1371/journal.pmed.1004303.t001

evaluation, the standard of care treatment was delivered using older WHO guidelines for FDCs available in the public health sector at the time [13]. In all 3 studies, study drugs were administered by the study staff on the day of pharmacokinetic sampling. More details about the studies included in this analysis are in Table 1.

Ethics statement

We pooled individual patient data from 3 multisite studies of children on the 2010 WHO-recommended first-line tuberculosis-drugs, summarized in Table 1. The DATiC study protocol was approved by the Health Research Ethics Committees of the Universities of Stellenbosch and Cape Town and the College of Medicine Research Ethics Committee in Malawi. The DNDi study protocol was approved by the ethics committees of Stellenbosch University, University of Cape Town, University of the Witwatersrand, and Pharma Ethics in Durban. The SHINE trial was approved by the Ethics Committees of the all participating sites in South Africa (Stellenbosch University and University of Cape Town Research Ethics Committees), Uganda (Joint Clinical Research Centre Institutional Review Board), Zambia (University of Zambia Biomedical Research Ethics Committee and National Health Research Ethics Committee), India (BJMC Ethics Committee-Pune, National Institute of Research in Tuberculosis Ethics Committee-Chennai and Health Ministry Screening Committee), and by University College London Research Ethics Committee. For all 3 studies, parents or legal guardians provided written informed consent and older children provided written assent.

Study sample analyses

Rifampicin, isoniazid, and pyrazinamide plasma concentrations were quantified using validated liquid chromatography with tandem mass spectrometry (LC-MS/MS) methods at University of Cape Town (UCT) Division of Clinical Pharmacology, with the exception of SHINE samples from India (5% of all samples), which were assayed using validated LC-MS/MS at National Institute of Research in Tuberculosis in India [14]. Three India-based participants' drug concentrations (21 samples) were assayed at both UCT and India for comparison. We adjusted for the overall difference in analysis methods of the 2 laboratories during pharmacokinetic model development. Assays at the UCT laboratory had a lower limit of quantification (LLOQ) of 0.117 mg/L for rifampicin, 0.0977 mg/L for isoniazid, and 0.200 mg/L for pyrazinamide. The LLOQ was 0.250 mg/L for all 3 drugs at the laboratory in India.

Population pharmacokinetic analysis

We analyzed concentration-time data using nonlinear mixed-effects modeling with NON-MEM version 7.4.4 (ICON Development Solutions, Ellicott City, Maryland, United States of America). We applied first-order conditional estimation with interaction (FOCE-I) and stochastic approximation expectation-maximization (SAEM) to estimate population pharmacokinetic parameters. To visualize data, we utilized Perl-speaks-NONMEM (PsN version 4.7.0), Pirana version 2.9.9, and R version 3.6.3 [15].

Samples below LLOQ (BLQ) were handled as described previously [13,16]. In short, 50% of the LLOQ value was imputed for censored values. For the first censored value closest to the peak concentration, the additive error was inflated by LLOQ/2, and all trailing censored values were excluded from the model fit but retained for simulation-based diagnostics. In the SHINE (samples from India) and DATiC studies, values between the LLOQ and lower limit of detection (LOD) were available, and these were used as such. Data below LOD were handled as described above.

We tested one- and two-compartment disposition models with first-order absorption with either transit compartments or lag time. We explored first-order elimination for the 3 drugs as well as first-pass and saturable elimination for rifampicin.

We included between-subject variability (BSV) on disposition parameters and betweenoccasion variability (BOV) on absorption parameters assuming a log-normal distribution. We described residual unexplained variability with a combined additive and proportional error model. To account for body size differences, we allometrically scaled (with either weight or fatfree mass) clearances and volumes of distribution with exponents of 0.75 and 1, respectively [17]. We investigated the association of age with clearance and bioavailability for all drugs and employed prior information to estimate the maturation of isoniazid and rifampicin clearance [18]. After including weight and age in the model, we tested additional covariates based on physiological plausibility, range of covariate values in the dataset, and assessment of empirical Bayes estimates versus covariate plots. N-acetyltransferase 2 (NAT2) acetylator-status (important for isoniazid clearance [19]) was available for 142 participants in the DATiC study. We applied a mixture-model for participants with unknown acetylator-status [20], fixed the proportions of fast, intermediate, and slow metabolizers to values reported previously in a study with a bigger sample size, and estimated the respective clearances [19]. We maintained covariates reaching statistical significance of p < 0.05 on forward addition and p < 0.01 on backward elimination. *P*-values were derived using the likelihood ratio test.

We modeled the datasets sequentially, attempting to correct for any systematic differences between studies, as previously suggested [21]. We assessed performance of intermediate and final models with visual predictive checks (VPCs) and evaluated parameter precision using a sampling importance resampling (SIR) procedure.

Simulations

We used final models to simulate area under concentration-time curve from 0 to 24 h (AUC_{0-24h}) and maximum concentrations (C_{max}) in an in silico representative population of 110,000 African pediatric patients (50% female) with uniformly distributed weight [22]. We compared simulated values with values reported in adults [5,23,24]. The following scenarios were investigated: (i) 2010 WHO-recommended pediatric tuberculosis weight-band (4.0–7.9, 8.0–11.9, 12.0–15.9, 16.0–24.9, and 25.0–36.9 kg) doses implemented with the available FDC (rifampicin/isoniazid/pyrazinamide 75 mg/50 mg/150 mg) (Fig 1 and Figure A in S1 Text); (ii) adding 75 mg or 150 mg tablet of rifampicin to top-up to 2010 WHO-recommended pediatric tuberculosis treatment doses using the available FDC (Fig 2); (iii) first-line tuberculosis treatment doses using the FDC with ratios (rifampicin/isoniazid/pyrazinamide 120/30/135 mg) and weight-bands (3<6, 6<13, 13<20, 20<25 kg) recommended by Denti and colleagues (Table A in S1 Text and Figure B in S1 Text) [13]; (iv) 2010 WHO-recommended pediatric tuberculosis doses using weight-bands commonly used for HIV (3<6, 6<10, 10<15, 15<20, 20<25, 25<30, 30<40 kg) [25] (Text B in S1 Text, Tables B and C in S1 Text, Text C in S1 Text, and Figure C in S1 Text); and (v) compared participants on lopinavir/ritonavir versus those not on lopinavir/ritonavir (Figs 3 and 4).

Results

Study population

Data from 387 children (116 SHINE, 91 DNDi, and 180 DATiC) were available for analysis. The median (range) age and weight were 2.2 (0.2 to 15.0) years and 10.9 (3.2 to 59.3) kg, respectively. Approximately 47% were female and 151 (39.0%) were living with HIV, of whom 95.4% were on ART. NAT2 acetylator-status was available in 142 (35.8%) children from DATiC, 35 (25%), 81 (57%), and 26 (18%) were slow, intermediate, and fast-acetylators, respectively. Participant characteristics are summarized in Table 2. Overall, 2,625, 1,996, and 1,865 concentrations for rifampicin, isoniazid, and pyrazinamide, respectively, were available for analysis. Of these, 21%, 19.5%, and 2.8% of these were below the LLOQ for rifampicin, isoniazid, and pyrazinamide, respectively.

Population pharmacokinetic analysis

A one-compartment disposition model with saturable clearance from a liver compartment and transit absorption best described rifampicin data. A two-compartment disposition model with first-order elimination and transit absorption fit isoniazid data best. A one-compartment disposition model with first-order elimination and absorption with lag time best described pyrazinamide data. Final parameter estimates with 95% confidence intervals (CIs) are shown in Table 3.





https://doi.org/10.1371/journal.pmed.1004303.g001

Allometric scaling of clearance and volume improved the fit for all 3 drugs. The exponents of clearance and volume were fixed, and there were therefore no additional degrees of freedom, hence, no *p*-values could be computed. Fat-free mass (FFM)-based scaling was better than weight for pyrazinamide (change in objective function value (dOFV) = 3.18), and weight was preferred for rifampicin (dOFV = 1.42). For isoniazid, as there was a very small difference between FFM and weight (dOFV = 0.390), we maintained the weight. The fit for all drugs further improved after we added maturation of clearance (dOFV = 34.8; *p* < 0.001, 42.9; *p* < 0.001, and 14.8; *p* < 0.001), for rifampicin, isoniazid, and pyrazinamide, respectively. Bioavailability increased linearly with age, until adult bioavailability was reached at 27.7 months (95% CI [20.7 months, 37.0 months]; dOFV = 38.3; *p* < 0.001) and 26.5 months (95% CI [14.2 months, 41.0 months]; dOFV = 24.2; *p* < 0.001) months for rifampicin and isoniazid, respectively. Pyrazinamide bioavailability was unchanged with age. Fig.5 shows the change in clearance and bioavailability from birth for all 3 drugs.

In children on lopinavir/ritonavir, rifampicin clearance was 22% (95% CI [13%, 28%]; dOFV = 19.0; p < 0.001) lower, while pyrazinamide clearance was increased by 49% (95% CI [39%, 57%]; dOFV = 130.0; p < 0.001) compared to all other children in the analysis. Isoniazid





Fig 2. Simulated rifampicin exposures after adding 75 mg or 150 mg to the WHO recommended doses and weight bands. Simulated rifampicin area under the curve from time zero to 24 h (AUC₀₋₂₄) after dose and maximum concentrations (C_{max}) at steady state, versus body weight, with concentrations achieved after topping up the existing FDC formulation with 75 mg and 150 mg tablets of rifampicin alone in children weighing <25 kg and >25 kg, respectively, using the WHO-recommended doses and weight-bands. The maroon boxes represent patients <3 months old while the pink boxes represent patients \geq 3 months. The shaded area represents the range of the reported adult median C_{max} and AUC₀₋₂₄. The box indicates the interquartile range, while the whiskers denote the 2.5th and the 97.5th percentiles. FDC, fixed-dose combination; WHO, World Health Organization.

https://doi.org/10.1371/journal.pmed.1004303.g002



Fig 3. Concentration versus time plots for simulated steady-state drug concentrations. **When taken at the same time*. Simulations of children weighing 14 kg on first-line tuberculosis drugs with or without antiretrovirals. Plot A shows the rifampicin concentrations achieved by a child on lopinavir/ritonavir versus when they are not on lopinavir/ritonavir. Plot B shows the isoniazid concentrations

achieved by a child on lopinavir/ritonavir taken at the same time with isoniazid or on efavirenz, versus when they are not on any antiretrovirals. Plot C shows the pyrazinamide concentrations achieved by a child on lopinavir/ritonavir versus when they are not on lopinavir/ritonavir. The grey dashed lines represent the lower and upper boundaries of the maximum concentrations (C_{max}) range reported in adults. The children were dosed according to 2010 WHO-recommended pediatric tuberculosis doses. Figures D, E, F, and G in <u>S1 Text</u> show VPCs stratified by study and lopinavir/ritonavir. VPC, visual predictive check; WHO, World Health Organization.

https://doi.org/10.1371/journal.pmed.1004303.g003

bioavailability was 39% (95% CI [32%, 45%], dOFV = 50.0; p < 0.001) lower in children taking their lopinavir/ritonavir dose at the same time as isoniazid (but not when the administration was at least 2 h apart) and 37% (95% CI [22%, 52%]; dOFV = 17.0; p < 0.001) lower in children receiving efavirenz, compared to all other children in the analysis. Figure D in S1 Text panels 1, 2, and 3 show the model fit for children who are HIV-, HIV+ not yet on ART, and HIV + who took lopinavir/ritonavir 2 h before isoniazid, while panels 4 and 5 show the model fit for children with HIV who took lopinavir/ritonavir at the same time as isoniazid and those on efavirenz, respectively. The bioavailability in the first 3 strata is similar, negating the influence of HIV on the pharmacokinetics of isoniazid, the last 2 strata have lower exposures, showing that the difference is present due to lopinavir/ritonavir taken at the same time with isoniazid and efavirenz co-treatment.



Fig 4. Simulated tuberculosis drug exposures achieved in children on lopinavir/ritonavir vs. those not on lopinavir/ritonavir. **When taken at the same time.* Simulated rifampicin, isoniazid, and pyrazinamide AUC from time zero to 24 h in steady state (AUC_{0-24h}) from concentrations achieved with the WHO-recommended weight-based dosing with the existing fixed dose combination formulation for approximately 65,000 patients weighing 3 to 36.9 kg. The pink boxes represent patients on lopinavir/ritonavir while the maroon boxes represent patients not on lopinavir/ritonavir. The difference in pharmacokinetic parameters between the 2 groups is indicated in the text above each plot. The isoniazid box plots are further stratified into fast, intermediate, and slow NAT2 acetylator-status. The gray horizontal dashed lines represent the upper and lower boundaries of the reported adult median AUC_{0-24h} ranges (72.9 to 38.7, 26.3 to 11.6, and 429 to 233 mg*h/L) for rifampicin, isoniazid, and pyrazinamide, respectively. The boxes indicate the interquartile range, while the whiskers denote the 2.5th and the 97.5th percentiles. AUC, area under the curve; NAT2, N-acetyl-2-transferase; WHO, World Health Organization.

https://doi.org/10.1371/journal.pmed.1004303.g004

Table 2. Patient characteristics stratified by study.

		SHINE[11] (N = 116)	DNDI[12] (N = 91)	DATIC[13] (N = 180)	Total (N = 387)
Weight (kg)	Median (Min–Max)	14.2 (5.5–59.3)	8.8 (3.9–14.9)	10.9 (3.2–28.8)	10.9 (3.2–59.3)
Height (cm)	Median (Min–Max)	95.5 (53.7-176)	73 (55–101)	80.6 (49.9–135)	81.0 (49.9–176)
Sex: n (%)					
Female		55/116 (47.4%)	52/91 (57.1%)	74/180 (41.1%)	181/387 (46.8%)
Male		61/116 (52.6%)	39/91 (42.9%)	106/180 (58.9%)	206/387 (53.2%)
Age (years)	Median (Min–Max)	4.3 (0.3-15)	1.6 (0.3-6.8)	2 (0.2–11.9)	2.2 (0.2–15)
Drug administration method	: n (%)				
Unknown		0 (0%)	91/91 (100%)	0 (0%)	91/387 (23.5%)
Whole		51/116 (44%)	0 (0%)	43/180 (23.9%)	94/387 (24.3%)
Dispersed or crushed		63/116 (54.3%)	0 (0%)	59/180 (32.8%)	122/387 (31.5%)
Crushed tablet and syringe		0 (0%)	0 (0%)	38/180 (21.1%)	38/387 (9.8%)
Crushed tablet and nasogast	ric tube	2/116 (1.7%)	0 (0%)	40/180 (22.2%)	42/387 (10.9%)
Formulation: n (%)					
pHRZ (McLeod's)*		55/116 (47.4%)	N/A	N/A	55/387 (14.2%)
pHR (McLeod's)**		38/116 (32.8%)	N/A	N/A	38/387 (9.8%)
HRZ (McLeod's)***		14/116 (12.1%)	N/A	N/A	14/387 (3.6%)
HR (McLeod's)#		9/116 (7.8%)	N/A	N/A	9/387 (2.3%)
RH_60-60 (Rimactazid, San	doz, RSA)##	N/A	91/91 (100%)	N/A	91/387 (23.5%)
R (EREMFAT)###		N/A	N/A	115/180 (63.9%)	115/387 (29.7%)
R-CIN^		N/A	N/A	59/180 (32.8%)	59/387 (15.2%)
RH (Rimactazid, Norvatis, I	ndia)^^	N/A	N/A	6/180 (3.3%)	6/387 (1.6%)
HIV Status: n (%)					
Negative		81/116 (69.8%)	0 (0%)	150/180 (83.3%)	231/387 (59.7%)
Positive		35/116 (30.2%)	91/91 (100%)	25/180 (13.9%)	151/387 (39%)
Exposed through mum		0 (0%)	0 (0%)	4/180 (2.2%)	4/387 (1%)
Missing		0 (0%)	0 (0%)	1/180 (0.6%)	1/387 (0.3%)
Antiretroviral therapy: n (%)					
Not yet started		2/116 (5.7%)	0 (0%)	5/180 (17.2%)	7/387 (4.5%)
Efavirenz		15/116 (42.9%)	0 (0%)	6/180 (20.8%)	21/387 (13.5%)
Lopinavir/ritonavir		18/116 (51.4%)	91/91 (100%)	13/180 (44.8%)	122/387 (78.7%)
Nevirapine		0 (0%)	0 (0%)	5/180 (17.2%)	5/387 (3.3%)
Rifampicin dose (mg/kg)^^^^	Median (Min-Max)	14.4 (8.6–19.7)	12.5 (9.2–18.2)	15.5 (10.1–20.5)	14.0 (8.6–20.5)
Isoniazid dose (mg/kg)	Median (Min-Max)	9.6 (3.9–17.6)	12.4 (9.2–18.2)	12.0 (10.0-21.5)	11.7 (3.9–21.5)
Pyrazinamide dose (mg/kg)	Median (Min-Max)	28.2 (18.8-37.5)	29.8 (19.2-43.1)	34.1 (23.4-41.7)	32.1 (18.8-43.1)

*Pediatric formulation containing 75 mg rifampicin, 50 mg isoniazid, and 150 mg pyrazinamide.

**Pediatric formulation containing 75 mg rifampicin and 50 mg isoniazid.

***Adult formulation containing 150 mg rifampicin, 75 mg isoniazid, and 400 mg pyrazinamide.

[#]Adult formulation containing 150 mg rifampicin and 75 mg isoniazid.

^{##}Pediatric formulation containing 60 mg rifampicin and 60 mg isoniazid.

###EREMFAT rifampicin only (20 mg/ml) granulates for suspension.

[^]R-Cin rifampicin only 100 mg/5 ml suspension.

 $^{\wedge\wedge}$ Rimactazid pediatric formulation containing 100 mg rifampicin and 50 mg isoniazid.

^{^^^}Following 2010 WHO pediatric tuberculosis dosing guidelines. N/A = not applicable.

DNDi, Drugs for Neglected Diseases initiative; HIV, human immunodeficiency virus; WHO, World Health Organization.

https://doi.org/10.1371/journal.pmed.1004303.t002

The R-Cin rifampicin-only formulation taken in the DATiC study had 63% (95% CI [57%, 69%], dOFV = 20.0; p < 0.001) lower bioavailability compared to other rifampicin formulations pooled together. The McLeod's adult drug FDC formulations used in SHINE had 34% (95% CI [20%, 47%]; dOFV = 18.0; p < 0.001) lower isoniazid bioavailability than other isoniazid formulations pooled together. Table 3 shows parameter estimates and covariates for all 3 drugs and Figures D, E, F, and G in S1 Text show the final model fit for all drugs.

Simulation results

AUC and C_{max} simulations with 2010 WHO-recommended tuberculosis weight-band dosing implemented with available FDC had median rifampicin, isoniazid, and pyrazinamide AUCs and C_{max} above reported adult exposures for children younger than 3 months.

For children older than 3 months, median rifampicin AUCs were lower than reported adult AUCs in all weight-bands except the 16.0 to 24.9 kg weight-band. For isoniazid, median AUCs were above observed adult AUCs in children weighing 12.0 to 24.9 kg. Median pyrazinamide AUCs were within observed adult AUCs in all weight-bands (Fig 1 and Figure A in S1 Text).

Simulations with additional rifampicin doses of 75 mg and 150 mg in children weighing <25 kg and >25 kg respectively, to top-up 2010 WHO-recommended weight-based tuberculosis doses, showed improved rifampicin exposures in children in all weight bands, leading to median exposures similar to those observed in adults (Fig 2).

Simulations with 2010 WHO-recommended tuberculosis doses implemented with the available FDC and weight-bands commonly used in HIV showed improved rifampicin exposures in children weighing 6 to 8, 10 to 12, and 14 to 16 kg (Tables B and C in <u>S1 Text</u> and Figure C in <u>S1 Text</u>), compared to using 2010 WHO-recommended weight-bands. Simulations with the new FDC and weight-bands suggested by Denti and colleagues showed the optimal exposures (Table A in <u>S1 Text</u> and Figure B in <u>S1 Text</u>) [13].

Simulations of scenarios with and without lopinavir/ritonavir showed increased rifampicin exposures in the former. The exposure-reducing impact of taking lopinavir/ritonavir at the same time with isoniazid was most pronounced in fast-acetylators. Lopinavir/ritonavir reduced pyrazinamide exposures (Figs 3 and 4).

Discussion

In this model-based pooled individual patient data pharmacokinetic analysis, we combined data from 3 studies of children on the 2010 WHO-recommended first-line tuberculosis treatment. This gave us a strong and unique dataset that allowed us to develop robust population pharmacokinetic models to evaluate the WHO dosing guidelines and explore the associations of HIV, ART, drug formulation, age, and body size with the pharmacokinetics of first-line tuberculosis drugs. We observed no association of HIV with the pharmacokinetics of rifampicin, isoniazid, or pyrazinamide. We found children on lopinavir/ritonavir had higher rifampicin, lower pyrazinamide, and lower isoniazid (when isoniazid and lopinavir/ritonavir were ingested simultaneously) exposures. Children on efavirenz also had lower isoniazid exposure. Two formulations led to further lower rifampicin and isoniazid exposures: the R-Cin rifampicin formulation and the Mcleod's adult FDC formulation, respectively.

There were significant drug-drug interactions between first-line tuberculosis drugs and lopinavir/ritonavir. There was 22% reduction in rifampicin clearance in children on lopinavir/ ritonavir compared to other children on rifampicin without lopinavir/ritonavir. Ritonavir inhibits intestinal and hepatic p-glycoprotein and also organic anion-transporting polypep-tides 1 and 3 efflux transporters in the liver, possibly resulting in reduced excretion of rifampicin into bile [26]. Our findings are consistent with 2 studies of adults on steady-state

Typical values (95% CI) ⁺								
Parameter (units)	Rifampicin	Isoniazid	Pyrazinamide					
Clearance-CL (L/h) ¹	52.5 (46.7, 59.0)		1.27 (1.19, 1.33)					
Slow Intermediate Fast	-	4.01 (3.61, 4.61)* 6.16 (5.46, 6.85)** 7.06 (6.33, 7.91)***	-					
Central volume of distribution—V (L)1	13.7 (12.7, 14.9)	13.8 (12.9, 14.8)	10.8 (10.4, 11.2)					
Intercompartmental clearance—Q (L/h)1	-	0.509 (0.438, 0.584)	-					
Peripheral volume of distribution (L)1	-	5.73 (4.85, 6.71)	_					
Bioavailability-F	1 FIXED	1 FIXED	1 FIXED					
Mean absorption transit time—MTT (h)	0.515 (0.439, 0.603)	0.300 (0.245, 0.362)	-					
Number of absorption transit compartments—NN (n)	18.5 (15.4, 22.3)	10 FIXED	-					
Lag time (h)	-	-	0.154 (0.098, 0.218)					
First-order absorption rate constant—Ka (1/h)	2.72 (1.72, 2.29)	3.56 (3.03, 3.79)	1.89 (1.65, 2.25)					
Hepatic blood flow—Q _H (L/h) ²	90 FIXED	-	_					
Hepatic volume $-V_{\rm H}$ (L) ²	1 FIXED	-	-					
Fraction of unbound drug—F _u	0.2 FIXED	-	-					
Michaelis-Menten rate constant K _m —(mg/L)	9.39 (7.96, 11.1)	-	-					
Formulation on F (%) ³	-63.3 (-57.4, -69.2); p < 0.001	-33.9 (-20.0, -46.5); $p < 0.001$	-					
Crushing tablets on Ka (%) ⁴	-	-	+98.0 (+64.0, +137); <i>p</i> < 0.001					
Lopinavir/ritonavir on CL (%)	-22.3 (-12.8 , -28.0); $p < 0.001$	-	+48.5 (+39.4, +57.7); <i>p</i> < 0.001					
Lopinavir/ritonavir on F (if ingested simultaneously) (%)	-	-39.2 (-32.1, -45.2); p < 0.001	-					
Efavirenz on F (%)	-	-37.2 (-21.8, -51.9); p < 0.001	-					
Yesterday's F (%)	-	-	-36.6(-29.3, -42); p < 0.001					
Study on F (%) ⁵	-	-	-27.2 (-20.9 , -34.1); $p < 0.001$					
Maturation of CL								
Post-menstrual age at 50% adult CL (months) ⁶	11.0 (9.90, 12.3); <i>p</i> < 0.001	10.9 (10.1, 11.7); <i>p</i> < 0.001	11.5 (10.2, 12.2); <i>p</i> < 0.001					
Gamma (y- shape of sigmoidal maturation function)	3.25 (2.72, 3.89); <i>p</i> < 0.001	3.31 (2.86, 3.95); <i>p</i> < 0.001	3.42 (2.29, 5.11); $p < 0.001$					
	Age on F							
F at Birth (%)	48.9 (36.0, 61.8); <i>p</i> < 0.001	64.5 (52.1, 78.9); <i>p</i> < 0.001	-					
Age at adult F (months)	27.7 (20.7, 37.0); <i>p</i> < 0.001	26.5 (14.2, 41.0); <i>p</i> < 0.001	-					
Lab_Scaling (fold change) ⁷	1.22 (0.933, 1.47); p < 0.001	-	1.34 (1.28, 1.42); <i>p</i> < 0.001					
	BSV (%)							
V	-	21.5 (17.0, 25.1)	-					
CL	34.9 (30.1, 39.2)	27.3 (23.2, 31.5)	22.0 (20.4, 24.5)					
BOV (%)								
F	35.4 (32.9, 37.6)	44.8 (40.9, 52.0)	29.8 (28.0, 33.2)					
MTT	42.1 (33.1, 49.4)	81.2 (70.3, 94.6)	-					
Ка	162 (143, 180)	82.0 (65.0, 98.2)	71.3 (61.8, 82.6)					
Lag time	-	-	91.7 (70.5, 121.6)					
Study or site on BOV on F (fold change)	-	1.76 (1.45, 2.11); $p < 0.001$	2.74 (2.18, 3.32); $p < 0.001$					
Residual unexplained error								
Proportional error (%)	24.1 (23.3, 24.9)	14.2 (13.5, 14.9)	7.21 (7.15, 8.03)					
Additive error (mg/L)	0.023 FIXED	0.0195 FIXED	0.0400 FIXED					

Table 3. Final model parameters for rifampicin, isoniazid, and pyrazinamide.

(Continued)

Table 3. (Continued)

Pre-dose additive error (mg/L) ⁸	_	-	1.10 (0.885, 1.25); <i>p</i> < 0.001
---	---	---	--------------------------------------

¹All clearances and volumes of distribution were allometrically scaled and the typical values reported here refer to a child weighing 14 kg. For rifampicin, the value reported is the intrinsic clearance.

²These values are for an adult male of 70 kg. Then, it was allometrically scaled by weight.

³This relationship was for the R-Cin formulation for rifampicin and the McLeod's adult formulation for isoniazid.

⁴Those that had their tablets crushed had a faster absorption.

⁵This was a scaling factor for the DNDi study.

⁶Post-natal age + gestation.

⁷This was a factor scaling the samples that were assayed in India within the SHINE study.

⁸Extra additive error on the pre-dose samples.

⁺Based on SIR.

*Proportion of slow NAT2 acetylators fixed to 44%.

**Proportion of intermediate NAT2 acetylators fixed to 42%.

***Proportion of fast NAT2 acetylators fixed to 14%.

^bThis relationship was for the DNDi study.

[#]This was a scaling factor for the Tygerberg site in the DNDi study.

BOV, between-occasion variability; BSV, between-subject variability; CI, confidence interval; DNDi, Drugs for Neglected Diseases initiative; NAT2, N-acetyltransferase 2; SIR, sampling importance resampling.

https://doi.org/10.1371/journal.pmed.1004303.t003

rifampicin where 5 patients coinfected with HIV and tuberculosis were on lopinavir/ritonavir and 10 healthy volunteers were on atazanavir [7,27]. Future studies considering higher rifampicin doses should pay attention to children on lopinavir/ritonavir co-treatment, as they may exhibit higher than desired rifampicin concentrations.

In children on lopinavir/ritonavir, pyrazinamide exposure was 32.2% lower due to increased clearance. In vitro evidence suggests this might be attributed to ritonavir's activation



Fig 5. Plot of maturation of clearance and bioavailability. Plot of maturation of clearance (solid line) and bioavailability (dashed line) of rifampicin, isoniazid, and pyrazinamide. The vertical lines at the bottom indicate the ages for which we had observations.

https://doi.org/10.1371/journal.pmed.1004303.g005

of xanthine oxidase [28], the enzyme which metabolizes pyrazinamide [29]. To our knowledge, this association has not been reported previously [7]. It is possible that we were able to detect it because 80% of the children on lopinavir/ritonavir were super-boosted by additional ritonavir (lopinavir/ritonavir 4:4 rather than 4:1) in the presence of rifampicin. Because pyrazinamide is essential in reducing tuberculosis relapse risk [30], the potential consequences of the reduced pyrazinamide exposure should be considered in designing ART regimens for children with tuberculosis.

When isoniazid and lopinavir/ritonavir were ingested simultaneously, lower isoniazid exposures were observed due to 39% reduced isoniazid bioavailability. We hypothesize that this is due to the high fructose content (15.3% w/v) in Kaletra, the lopinavir/ritonavir formulation used in all 3 studies [31]. Rao and colleagues noted isoniazid condensation when mixed with different sugars [32,33]. We hypothesize that an interaction between fructose and isoniazid may have led to reduced isoniazid available for absorption. Children on isoniazid-containing tuberculosis treatment may benefit if ingestion of co-prescribed syrups is separated by at least 1 or 2 h from isoniazid.

The association of efavirenz with isoniazid pharmacokinetics has been described previously [6,34]. Although these studies reported an increased clearance of isoniazid in fast-acetylating adults, our finding of lower isoniazid exposure in children on efavirenz was independent of isoniazid acetylator-status. This is probably due to the small number (13.7%) of children on efavirenz in our study, and missing NAT2 acetylator-status information in some children, which limited our ability to detect the explanatory effect of acetylator-status. Additionally, in children in this analysis, efavirenz was administered within the same time frame as isoniazid. The association we found could be a result of pre-hepatic interactions between efavirenz and isoniazid, but we cannot make conclusions since we did not have a group which took efavirenz hours apart from isoniazid, to serve as a comparator.

Since the associations of efavirenz and lopinavir/ritonavir on isoniazid bioavailability were similar, we explored a possible HIV association. We had 7 children with HIV, not yet on ART, and we could separate the association of each ART and HIV. We found no difference in isoniazid bioavailability among those without HIV and those with HIV but ART-naive, ruling out the association of HIV with isoniazid bioavailability (Figure D in S1 Text) [7]. Similar results were found when repeating the same test for rifampicin and pyrazinamide, where HIV infection did not change their pharmacokinetics. Most previous studies evaluating the association of HIV with tuberculosis-drug pharmacokinetics in children did so in the presence of ART [14,35,36]; hence, it may have been impossible to distinguish whether any association was due to HIV or to ART. Two studies that recruited ART-naïve children reported that HIV reduced the exposure of rifampicin, pyrazinamide, and ethambutol [37,38]. However, both these studies used non-compartmental analysis, a methodology with limited ability to account for confounding factors and nonlinear covariates (like age and weight), so the reported association of HIV may have been confounded by dose and malnutrition, which likely overlapped with HIV in these studies. Our analysis used nonlinear mixed effect modeling, and we could better account for the nonlinear confounding-factors.

With regards to formulation, we confirmed 63% lower rifampicin bioavailability in the R-Cin formulation compared to other rifampicin formulations, as reported by Denti and colleagues [13]. This is worrying because it exacerbates the problem of rifampicin under-exposure, especially in children weighing less than 12 kg, whom we found to have very low rifampicin AUC compared to those reported in adults. We also confirmed previous findings of lower AUC of isoniazid in children on the adult formulation in SHINE [39]. Furthermore, in our analysis we were able to conclude that this reduction was not only due to the lower dose for children in the >25 kg weight-band (4 to 6 mg/kg versus 10 mg/kg) or allometry, since

both factors were accounted for in the model. These results stress the importance of continued drug-quality monitoring and bioequivalence studies in intended populations.

Results from our simulations showed that the 2010 WHO-recommended weight-based dosing of first-line tuberculosis-drugs leads to suboptimal drug exposures because it does not account for young children's maturation of clearance and does not completely account for the differences in body size between children and adults [40]. Of note, children under 3 months were predicted to have higher median AUC_{0-24h} and C_{max} for all 3 drugs, when compared to observed adult exposures. Although we did not have many children under 3 months, our results are consistent with previous reports [13]. Whether higher exposures in this age-group lead to drug-induced toxicities remains to be investigated. Moreover, the safe and effective range for these medicines could be wider than the reported observed adult median values. As a short-term precautionary measure, we recommend monitoring for hepatic toxicity symptoms in children younger than 3 months during first-line tuberculosis treatment. In children older than 3 months, median rifampicin exposures were low in most weight-bands, confirming previous findings [13,41]. We recommend higher rifampicin doses in children older than 3 months. As a quick, short-term solution, we suggest topping up rifampicin doses with 75 mg and 150 mg rifampicin only tablets in children weighing <25 kg and >25 kg, respectively. A rifampicin top-up would be feasible to implement short-term because the existing 150 mg single drug rifampicin formulations could be used (half a tablet for children weighing <25 kg) and no new FDC would have to be manufactured. As a long-term solution, we recommend adopting the doses, FDC and weight-bands recommended by Denti and colleagues [13] as shown in Figure B in S1 Text. Finally, if the WHO endorses harmonizing weight-bands across different diseases using weight bands commonly used in HIV (using 2010 WHO-recommended doses and currently available FDCs), the exposures achieved would be comparable with those achieved with the 2010 WHO-recommended weight-bands, with slightly increased exposures in children weighing 6 to 16 kg. With these weight bands, we also recommend supplementing rifampicin with 75 mg and 150 mg rifampicin only tablets in children weighing <25 kg and >25 kg, respectively. Using the HIV weight-bands would benefit HIV and tuberculosis coinfected patients and healthcare prescribers who would find it convenient to use the same weight-bands. Increased rifampicin doses could exacerbate drug-drug interactions and might warrant re-evaluation, but recent findings have shown that the additional induction effect is limited [41].

The strengths of our study are that firstly, we pooled 3 large studies conducted almost synchronously, all intensively sampled, and whose drug concentrations for the majority were assayed in the same laboratory, which substantially reduced impact of inaccuracy in drug quantification [42]. This large and unique dataset enabled us to develop robust models exploring and accounting for associations of age and body size, drug-formulation, HIV, and ART. Our analysis has an advantage over pooling individual summary measures, which accumulate errors and/or bias contained in the calculations derived differently for different studies [43]. The main limitation of our analysis was that, despite the huge similarities in the studies and consistency in the assay method across studies, there were small pharmacokinetic differences that had to be accounted for in the model. For example, all children in DNDi were on superboosted lopinavir/ritonavir while only a few were on this ART regimen in other studies, different tuberculosis drug formulations were used across the studies, some children in the DATiC study were given lopinavir/ritonavir simultaneously with isoniazid while others were not. This is likely the reason why we still had study and site-specific differences that could not be explained otherwise. For this reason, future studies need to independently confirm our findings, since some unaccounted confounders cannot be ruled out. This also calls for standardization of pediatric pharmacokinetic studies and collection of accurate information about drug

formulation, method, and timing of administration of all drugs and food intake. This would make future pooling of studies easier. Finally, having NAT2 genotype information in the whole population may have reduced this unexplained study variability and further reduced risks of confounding by study.

In conclusion, the main finding of our study is that children older than 3 months on the 2010 WHO-recommended dosing achieve lower rifampicin exposures than adults. On the other hand, rifampicin, isoniazid, and pyrazinamide drug concentrations may be too high in children younger than 3 months, for whom we recommend toxicity monitoring. A further revision to WHO pediatric tuberculosis dosing guidelines could address these issues. Increasing rifampicin doses would be a good start and can be quickly achieved by adding a rifampicin tablet. If harmonized weight bands are adopted, then higher rifampicin doses would still apply. If the guidelines were updated completely, we would recommend the optimized FDC formulation and optimized weight-bands recommended by Denti and colleagues [13]. In terms of associations of HIV and ART, we report that drug–drug interactions with ART, and not HIV infection per se [14,35,36], may alter tuberculosis-drug concentrations. As a precaution, we recommend that where tuberculosis-drugs and sugar-containing ART formulations are concurrently indicated, if possible, they should be taken at least 2 h apart to avoid possible prehepatic interactions. Finally, we advocate stricter monitoring of tuberculosis drug formulations.

Supporting information

S1 Text. Supplementary material. Figure A. Simulated drug exposures after administration of WHO recommended doses and weight bands. Text A. DATiC recommended optimized weight bands and fixed dose combination tablets. Table A. DATiC recommended optimized weight bands and fixed dose combination tablet compared with WHO recommended doses and weight bands. Figure B. Simulated drug exposures achieved after administration of DATiC recommend fixed dose combination tablets and weight bands. Text B. Harmonized weight bands. Table B. Number of tablets to be taken once daily based on WHO-recommended doses, the current fixed-dose combination formulation* and harmonized weight bands. Table C. Number of tablets to be taken once daily by children weighing >25 kg based on WHO recommended adult doses, the current adult fixed-dose combination formulation and harmonized weight bands. Text C. Harmonized weight band simulation results. Figure C. Simulated drug exposures achieved after administration of WHO-recommended doses with harmonized weight bands (similar to those used in HIV). Figure D. Visual predictive check of isoniazid concentration versus time after dose, stratified by HIV status and antiretroviral therapy. Figure E. Visual predictive check of drug concentration versus time after dose, stratified by study. Figure F. Visual predictive check of rifampicin drug concentration versus time after dose, stratified by lopinavir/ritonavir. Figure G. Visual predictive check of pyrazinamide drug concentration versus time after dose, stratified by lopinavir/ritonavir. (DOCX)

Acknowledgments

We are thankful to the study participants and their families. We also thank the ICTS High Performance Computing team at the University of Cape Town (http://hpc.uct.ac.za) for providing us with the resources to perform the calculations in this study. The Drugs for Neglected Diseases *initiative* (DND*i*) is grateful to its donors, public and private, who have provided funding to DND*i* since its inception in 2003. A full list of DND*i*'s donors can be found at <u>www.dndi</u>. org/about/donors.

Author Contributions

Conceptualization: Helen McIlleron, Paolo Denti.

Data curation: Lufina Tsirizani Galileya.

Formal analysis: Lufina Tsirizani Galileya.

Investigation: Chishala Chabala, Helena Rabie, Janice Lee, Irene Njahira Mukui, Anneke Hesseling, Heather Zar, Rob Aarnoutse, Anna Turkova, Diana Gibb, Mark F. Cotton, Helen McIlleron.

Methodology: Roeland E. Wasmann, Helen McIlleron, Paolo Denti.

Supervision: Roeland E. Wasmann, Helen McIlleron, Paolo Denti.

Writing - original draft: Lufina Tsirizani Galileya.

Writing – review & editing: Roeland E. Wasmann, Chishala Chabala, Helena Rabie, Janice Lee, Irene Njahira Mukui, Anneke Hesseling, Heather Zar, Rob Aarnoutse, Anna Turkova, Diana Gibb, Mark F. Cotton, Helen McIlleron, Paolo Denti.

References

- World Health Organization (WHO). Global Tuberculosis Report 2021. 2021 May 19; [accessed 2021 Nov 22]. Available from: https://www.who.int/publications/i/item/9789240037021.
- 2. Lamb GS, Starke JR. Tuberculosis in Infants and Children. Microbiol Spectra. 2017 Mar 10; 5(2):553–7. https://doi.org/10.1128/microbiolspec.TNMI7-0037-2016 PMID: 28387193
- Getahun H, Gunneberg C, Granich R, Nunn P. HIV Infection–Associated Tuberculosis: The Epidemiology and the Response. Clin Infect Dis. 2010 May 15; 50(s3):S201–7. <u>https://doi.org/10.1086/651492</u> PMID: 20397949
- 4. World Health Organization (WHO). Rapid advice: treatment of tuberculosis in children. 2010; [accessed 2021 May 17]. Available from: https://apps.who.int/iris/handle/10665/44444.
- Daskapan A, Idrus LR, Postma MJ, Wilffert B, Kosterink JGW, Stienstra Y, et al. A Systematic Review on the Effect of HIV Infection on the Pharmacokinetics of First-Line Tuberculosis Drugs. Clin Pharmacokinet. 2019 Jun 8; 58(6):747–66. https://doi.org/10.1007/s40262-018-0716-8 PMID: 30406475
- Sekaggya-Wiltshire C, Chirehwa M, Musaazi J, von Braun A, Buzibye A, Muller D, et al. Low Antituberculosis Drug Concentrations in HIV-Tuberculosis-Coinfected Adults with Low Body Weight: Is It Time To Update Dosing Guidelines? Antimicrob Agents Chemother. 2019 Jun; 1:63(6). https://doi.org/10. 1128/AAC.02174-18 PMID: 30910890
- Rockwood N, Meintjes G, Chirehwa M, Wiesner L, McIlleron H, Wilkinson RJ, et al. HIV-1 Coinfection Does Not Reduce Exposure to Rifampin, Isoniazid, and Pyrazinamide in South African Tuberculosis Outpatients. Antimicrob Agents Chemother. 2016 Oct; 60(10):6050–9. <u>https://doi.org/10.1128/AAC.</u> 00480-16 PMID: 27480859
- Kim ES, Bruinooge SS, Roberts S, Ison G, Lin NU, Gore L, et al. Broadening Eligibility Criteria to Make Clinical Trials More Representative: American Society of Clinical Oncology and Friends of Cancer Research Joint Research Statement. J Clin Oncol. 2017 Nov 20; 35(33):3737–44. https://doi.org/10. 1200/JCO.2017.73.7916 PMID: 28968170
- Dheda K, Gumbo T, Gandhi NR, Murray M, Theron G, Udwadia Z, et al. Global control of tuberculosis: from extensively drug-resistant to untreatable tuberculosis. Lancet Respir Med. 2014 Apr; 2(4):321–38. https://doi.org/10.1016/S2213-2600(14)70031-1 PMID: 24717628
- Forget EJ, Menzies D. Adverse reactions to first-line antituberculosis drugs. Expert Opin Drug Saf. 2006 Mar 27; 5(2):231–49. https://doi.org/10.1517/14740338.5.2.231 PMID: 16503745
- Chabala C, Turkova A, Thomason MJ, Wobudeya E, Hissar S, Mave V, et al. Shorter treatment for minimal tuberculosis (TB) in children (SHINE): a study protocol for a randomised controlled trial. Trials. 2018 Dec 19; 19(1):237. https://doi.org/10.1186/s13063-018-2608-5 PMID: 29673395
- Rabie H, Denti P, Lee J, Masango M, Coovadia A, Pillay S, et al. Lopinavir–ritonavir super-boosting in young HIV-infected children on rifampicin-based tuberculosis therapy compared with lopinavir–ritonavir without rifampicin: a pharmacokinetic modelling and clinical study. Lancet HIV. 2019 Jan; 6(1):e32–42. https://doi.org/10.1016/S2352-3018(18)30293-5 PMID: 30529029

- Denti P, Wasmann RE, van Rie A, Winckler J, Bekker A, Rabie H, et al. Optimizing Dosing and Fixed-Dose Combinations of Rifampicin, Isoniazid, and Pyrazinamide in Pediatric Patients With Tuberculosis: A Prospective Population Pharmacokinetic Study. Clin Infect Dis. 2022 Aug 24; 75(1):141–51. https:// doi.org/10.1093/cid/ciab908 PMID: 34665866
- Bekker A, Schaaf HS, Draper HR, van der Laan L, Murray S, Wiesner L, et al. Pharmacokinetics of Rifampin, Isoniazid, Pyrazinamide, and Ethambutol in Infants Dosed According to Revised WHO-Recommended Treatment Guidelines. Antimicrob Agents Chemother. 2016 Apr; 60(4):2171–9. https://doi. org/10.1128/AAC.02600-15 PMID: 26810651
- Keizer R, Karlsson M, Hooker A. Modeling and Simulation Workbench for NONMEM: Tutorial on Pirana, PsN, and Xpose. CPT Pharmacometrics Syst Pharmacol. 2013 Jun; 2(6):50. <u>https://doi.org/10. 1038/psp.2013.24</u> PMID: 23836189
- Bergstrand M, Karlsson MO. Handling Data Below the Limit of Quantification in Mixed Effect Models. AAPS J. 2009 Jun 19; 11(2):371–80. https://doi.org/10.1208/s12248-009-9112-5 PMID: 19452283
- Al-Sallami HS, Goulding A, Grant A, Taylor R, Holford N, Duffull SB. Prediction of Fat-Free Mass in Children. Clin Pharmacokinet. 2015 Nov 5; 54(11):1169–78. <u>https://doi.org/10.1007/s40262-015-0277-z</u> PMID: 25940825
- Gisleskog PO, Karlsson MO, Beal SL. Use of prior information to stabilize a population data analysis. J Pharmacokinet Pharmacodyn. 2002 Dec; 29(5–6):473–505. https://doi.org/10.1023/a:1022972420004 PMID: 12795242
- Gausi K, Wiesner L, Norman J, Wallis CL, Onyango-Makumbi C, Chipato T, et al. Pharmacokinetics and Drug-Drug Interactions of Isoniazid and Efavirenz in Pregnant Women Living With HIV in High TB Incidence Settings: Importance of Genotyping. Clin Pharmacol Ther. 2021 Apr 16; 109(4):1034–44. https://doi.org/10.1002/cpt.2044 PMID: 32909316
- Keizer RJ, Zandvliet AS, Beijnen JH, Schellens JHM, Huitema ADR. Performance of Methods for Handling Missing Categorical Covariate Data in Population Pharmacokinetic Analyses. AAPS J. 2012 Sep; 14(3):601–11. https://doi.org/10.1208/s12248-012-9373-2 PMID: 22648902
- Svensson E, van der Walt J-S, Barnes KI, Cohen K, Kredo T, Huitema A, et al. Integration of data from multiple sources for simultaneous modelling analysis: experience from nevirapine population pharmacokinetics. Br J Clin Pharmacol. 2012 Sep; 74(3):465–76. https://doi.org/10.1111/j.1365-2125.2012. 04205.x PMID: 22300396
- Wasmann RE, Svensson EM, Walker AS, Clements MN, Denti P. Constructing a representative in-silico population for paediatric simulations: Application to HIV-positive African children. Br J Clin Pharmacol. 2021 Jul 11; 87(7):2847–54. https://doi.org/10.1111/bcp.14694 PMID: 33294979
- Stott KE, Pertinez H, Sturkenboom MGG, Boeree MJ, Aarnoutse R, Ramachandran G, et al. Pharmacokinetics of rifampicin in adult TB patients and healthy volunteers: a systematic review and meta-analysis. J Antimicrob Chemother. 2018 Sep 1; 73(9):2305–13. https://doi.org/10.1093/jac/dky152 PMID: 29701775
- Alsultan A, Peloquin CA. Therapeutic Drug Monitoring in the Treatment of Tuberculosis: An Update. Drugs. 2014 Jun 21; 74(8):839–54. https://doi.org/10.1007/s40265-014-0222-8 PMID: 24846578
- World Health Organization. Consolidated guidelines on HIV prevention, testing, treatment, service delivery and monitoring: recommendations for a public health approach. 2021. [accessed 2022 Sep 19]; p. 548. Available from: https://www.who.int/publications/i/item/9789240031593.
- Niemi M, Pasanen MK, Neuvonen PJ. Organic Anion Transporting Polypeptide 1B1: a Genetically Polymorphic Transporter of Major Importance for Hepatic Drug Uptake. Pharmacol Rev. 2011 Mar; 63 (1):157–81. https://doi.org/10.1124/pr.110.002857 PMID: 21245207
- Acosta EP, Kendall MA, Gerber JG, Alston-Smith B, Koletar SL, Zolopa AR, et al. Effect of Concomitantly Administered Rifampin on the Pharmacokinetics and Safety of Atazanavir Administered Twice Daily. Antimicrob Agents Chemother. 2007 Sep; 51(9):3104–10. https://doi.org/10.1128/AAC.00341-07 PMID: 17576825
- Lü J-M, Jiang J, Jamaluddin M, Liang Z, Yao Q, Chen C. Ginsenoside Rb1 Blocks Ritonavir-Induced Oxidative Stress and eNOS Downregulation through Activation of Estrogen Receptor-Beta and Upregulation of SOD in Human Endothelial Cells. Int J Mol Sci. 2019 Jan 12; 20(2):294. https://doi.org/10. 3390/ijms20020294 PMID: 30642080
- Sarkar S, Ganguly A. Current Overview of Anti-Tuberculosis Drugs: Metabolism and Toxicities. Mycobact Dis. 2016; 6(2). https://doi.org/10.4172/2161-1068.1000209
- Njire M, Tan Y, Mugweru J, Wang C, Guo J, Yew W, et al. Pyrazinamide resistance in Mycobacterium tuberculosis: Review and update. Adv Med Sci. 2016 Mar; 61(1):63–71. https://doi.org/10.1016/j. advms.2015.09.007 PMID: 26521205
- AbbVie Pharmaceutical Research & Development. Kaletra 80mg/20mg Oral Solution. 2022; [accessed 2022 Mar 9]. Available from: https://www.medicines.org.uk/emc/product/1315/smpc#gref.

- Rao KV, Kailasam S, Menon NK, Radhakrishna S. Inactivation of isoniazid by condensation in a syrup preparation. Bull World Health Organ. 1971; 45(5):625–632. PMID: 5316955.
- Becker C, Dressman JB, Amidon GL, Junginger HE, Kopp S, Midha KK, et al. Biowaiver Monographs for Immediate Release Solid Oral Dosage Forms: Isoniazid**A project of the International Pharmaceutical Federation FIP, Groupe BCS, www.fip.org/bcs. J Pharm Sci. 2007 Mar 1; 96(3):522–31. https:// doi.org/10.1002/jps.20765 PMID: 17117431
- Chirehwa MT, McIlleron H, Wiesner L, Affolabi D, Bah-Sow O, Merle C, et al. Effect of efavirenz-based antiretroviral therapy and high-dose rifampicin on the pharmacokinetics of isoniazid and acetyl-isoniazid. J Antimicrob Chemother. 2018 Sep 18; 7800:7719. https://doi.org/10.1093/jac/dky378 PMID: 30239829
- Mukherjee A, Velpandian T, Singla M, Kanhiya K, Kabra SK, Lodha R. Pharmacokinetics of isoniazid, rifampicin, pyrazinamide and ethambutol in HIV-infected Indian children. Int J Tuberc Lung Dis. 2016 May 1; 20(5):666–72. https://doi.org/10.5588/ijtld.15.0288 PMID: 27084822
- Graham SM, Bell DJ, Nyirongo S, Hartkoorn R, Ward SA, Molyneux EM. Low Levels of Pyrazinamide and Ethambutol in Children with Tuberculosis and Impact of Age, Nutritional Status, and Human Immunodeficiency Virus Infection. Antimicrob Agents Chemother. 2006 Feb; 50(2):407–13. https://doi.org/ 10.1128/AAC.50.2.407-413.2006 PMID: 16436690
- Antwi S, Yang H, Enimil A, Sarfo AM, Gillani FS, Ansong D, et al. Pharmacokinetics of the First-Line Antituberculosis Drugs in Ghanaian Children with Tuberculosis with or without HIV Coinfection. Antimicrob Agents Chemother. 2017 Feb; 61(2). https://doi.org/10.1128/AAC.01701-16 PMID: 27855070
- Ramachandran G, Kumar AKH, Kannan T, Bhavani PK, Kumar SR, Gangadevi NP, et al. Low Serum Concentrations of Rifampicin and Pyrazinamide Associated with Poor Treatment Outcomes in Children with Tuberculosis Related to HIV Status. Pediatr Infect Dis J. 2016 May; 35(5):530–4. <u>https://doi.org/10.1097/INF.00000000001069</u> PMID: 26825153
- Chabala C, Turkova A, Hesseling AC, Zimba KM, van der Zalm M, Kapasa M, et al. Pharmacokinetics of First-Line Drugs in Children With Tuberculosis, Using World Health Organization–Recommended Weight Band Doses and Formulations. Clin Infect Dis. 2021 Aug 22. https://doi.org/10.1093/cid/ciab725 PMID: 34420049
- 40. Denti P, Wasmann RE, Francis J, McIlleron H, Sugandhi N, Cressey TR, et al. One dose does not fit all: revising the WHO paediatric dosing tool to include the non-linear effect of body size and maturation. Lancet Child Adolesc Health. 2022 Jan 1; 6(1):9–10. https://doi.org/10.1016/S2352-4642(21)00302-3 PMID: 34678142
- Garcia-Prats AJ, Svensson EM, Winckler J, Draper HR, Fairlie L, van der Laan LE, et al. Pharmacokinetics and safety of high-dose rifampicin in children with TB: the Opti-Rif trial. J Antimicrob Chemother. 2021 Nov 12; 76(12):3237–46. https://doi.org/10.1093/jac/dkab336 PMID: 34529779
- Aarnoutse RE, Sturkenboom MGG, Robijns K, Harteveld AR, Greijdanus B, Uges DRA, et al. An interlaboratory quality control programme for the measurement of tuberculosis drugs. Eur Respir J. 2015 Jul; 46(1):268–71. https://doi.org/10.1183/09031936.00177014 PMID: 25882800
- 43. Gafar F, Wasmann RE, McIlleron HM, Aarnoutse RE, Schaaf HS, Marais BJ, et al. Global estimates and determinants of antituberculosis drug pharmacokinetics in children and adolescents: a systematic review and individual patient data meta-analysis. Eur Respir J. 2023 Mar; 61(3):2201596. https://doi. org/10.1183/13993003.01596-2022 PMID: 36328357